Author's response to reviews

Title: Twenty-four hour metabolic rate measurements utilized as a reference to evaluate several prediction equations for calculating energy requirements in healthy infants

Authors:

Russell Rising (Russell_rising@yahoo.com)

Version: 5 Date: 1 February 2011

Author's response to reviews:

Just corrected one typo in the tables, one in the statistics section and one in the abstract.